

Citation:

Carrivick, JL and Boston, CM and Sutherland, JL and Pearce, D and Armstrong, H and Bjørk, A and Kjeldsen, KK and Abermann, J and Oien, RP and Grimes, M and James, WHM and Smith, MW (2023) Research Letter: Mass Loss of Glaciers and Ice Caps Across Greenland Since the Little Ice Age. Geophysical Research Letters, 50 (10). pp. 1-9. ISSN 0094-8276 DOI: https://doi.org/10.1029/2023gl103950

Link to Leeds Beckett Repository record: https://eprints.leedsbeckett.ac.uk/id/eprint/9729/

Document Version: Article (Published Version)

Creative Commons: Attribution 4.0

© 2023. The Authors.

The aim of the Leeds Beckett Repository is to provide open access to our research, as required by funder policies and permitted by publishers and copyright law.

The Leeds Beckett repository holds a wide range of publications, each of which has been checked for copyright and the relevant embargo period has been applied by the Research Services team.

We operate on a standard take-down policy. If you are the author or publisher of an output and you would like it removed from the repository, please contact us and we will investigate on a case-by-case basis.

Each thesis in the repository has been cleared where necessary by the author for third party copyright. If you would like a thesis to be removed from the repository or believe there is an issue with copyright, please contact us on openaccess@leedsbeckett.ac.uk and we will investigate on a case-by-case basis.

ᠭ



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL103950

Key Points:

- Total volume loss of at least 587 km³ since the Little Ice Age (LIA) termination, equating to 499 Gt and to 1.38 mm sea level equivalent
- Glacier mass balance from 2000 to 2019 is three times more negative than since the LIA but five times more negative in the North region
- Lake-terminating glaciers have experienced the greatest change in rate of mass loss

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. L. Carrivick, j.l.carrivick@leeds.ac.uk

Citation:

Carrivick, J. L., Boston, C. M., Sutherland, J. L., Pearce, D., Armstrong, H., Bjørk, A., et al. (2023). Mass loss of glaciers and ice caps across Greenland since the Little Ice Age. *Geophysical Research Letters*, 50, e2023GL103950. https://doi.org/10.1029/2023GL103950

Received 31 MAR 2023 Accepted 9 MAY 2023

Author Contributions:

Conceptualization: Jonathan L. Carrivick

Formal analysis: Jonathan L. Carrivick Investigation: Jonathan L. Carrivick, Clare M. Boston, Jenna L. Sutherland, Danni Pearce, Hugo Armstrong, Rachel P. Oien, Michael Grimes, Mark W. Smith Methodology: Jonathan L. Carrivick Project Administration: Jonathan L. Carrivick

Resources: Anders Bjørk, Kristian K. Kjeldsen, Jakob Abermann, William H. M. James

Visualization: Jonathan L. Carrivick Writing – original draft: Jonathan L. Carrivick

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Mass Loss of Glaciers and Ice Caps Across Greenland Since the Little Ice Age

Jonathan L. Carrivick¹, Clare M. Boston², Jenna L. Sutherland³, Danni Pearce⁴, Hugo Armstrong¹, Anders Bjørk⁵, Kristian K. Kjeldsen⁶, Jakob Abermann⁷, Rachel P. Oien⁸, Michael Grimes¹, William H. M. James¹, and Mark W. Smith¹

¹School of Geography and Water@Leeds, University of Leeds, Leeds, UK, ²School of the Environment, Geography and Geosciences, University of Portsmouth, Portsmouth, UK, ³School of Built Environment, Engineering and Computing, Leeds Beckett University, Leeds, UK, ⁴Faculty of Environmental Science and Natural Resource Management, Norwegian University of Life Sciences, Ås, Norway, ⁵Department of Geoscience and Natural Resources, University of Copenhagen, Copenhagen, Denmark, ⁶Department of Glaciology and Climate, Geological Survey of Denmark and Greenland, Copenhagen, Denmark, ⁷Department of Geography and Regional Science, University of Graz, Graz, Austria, ⁸Department of Geology, University at Buffalo, Buffalo, NY, USA

Abstract Glaciers and ice caps (GICs) are important contributors of meltwater runoff and to global sea level rise. However, knowledge of GIC mass changes is largely restricted to the last few decades. Here we show the extent of 5327 Greenland GICs during Little Ice Age (LIA) termination (1900) and reveal that they have fragmented into 5467 glaciers in 2001, losing at least 587 km³ from their ablation areas, equating to 499 Gt at a rate of 4.34 Gt yr⁻¹. We estimate that the long-term mean mass balance in glacier ablation areas has been at least -0.18 to -0.22 m w.e. yr⁻¹ and note the rate between 2000 and 2019 has been three times that. Glaciers with ice-marginal lakes formed since the LIA termination have had the fastest changing mass balance. Considerable spatial variability in glacier changes suggest compounding regional and local factors present challenges for understanding glacier evolution.

Plain Language Summary Glaciers and ice caps of Greenland peripheral to the ice sheet are important contributors of meltwater to the oceans and to global sea-level rise. In this study we map the extent of 5467 glaciers during the Little Ice Age (LIA) termination c. 1900 and calculate that they have lost at least 587 km³. The rate of mass change of these glaciers between 2000 and 2019 was three times more negative than the long-term average (of 4.34 Gt yr⁻¹) since the LIA. Lake-terminating glaciers now lose mass the fastest compared with land- or marine-terminating glaciers. Considerable spatial variability in glacier responses suggests local factors are important and makes glacier evolution complex.

1. Introduction

The Arctic has warmed nearly four times faster than the global average since the 1970s (Rantanen et al., 2022). In the short-term, this pronounced, amplified warming most severely impacts the glaciers and ice caps (GICs) of Greenland because they have quicker response times than the Greenland Ice Sheet (GrIS) (Khan et al., 2022; Larsen et al., 2022; Noël et al., 2017).

Greenland GICs constitute between $\sim 8.5\%$ and 10% of the world's total GIC volume, or $\sim 11\%$ to 16% excluding Antarctic Peninsula (Farinotti et al., 2019; Millan et al., 2022). They presently deliver the second largest contribution (13%) of meltwater from GICs to global sea level behind GICs in Alaska (Hugonnet et al., 2021).

Additionally, meltwater runoff from Greenland into the North Atlantic affects ocean circulation and hence the climate of western Europe, which has feedbacks on human health, behavior and economy (Siegert et al., 2020). Within Greenland, runoff affects fjord water quality and circulation, which has consequences for marine ecosystems and their primary production (Hopwood et al., 2020). Moreover, runoff impacts the Greenlandic population via fishing, mining, and hydropower (e.g., Sugiyama et al., 2021) as well as tourism.

GICs represent only $\sim 5\%$ of the glacier-covered area of Greenland and correspond to 26.8 mm sea level equivalent (SLE) of the volume of the GrIS (Millan et al., 2022). However, the mass loss of Greenland GICs has increased since the late 1990s, much more so than that of the GrIS (Noël et al., 2017). Overall, Greenland GICs could lose between 19% and 28% of their volume by 2100, reflecting regional differences in temperature and

precipitation (Machguth et al., 2013). However, satellite-derived data sets have revealed that there has been considerable variability in the absolute amount and rate of change of mass loss between Greenland regions (Hugonnet et al., 2021; Khan et al., 2022).

These recent decadal-scale changes to Greenland GICs lack a longer-term (pre-satellite era, centennial) context. A longer-term context is important because projections of Greenland GICs into the future require base line data sets, such as past glacier extents and past glacier ice surfaces, to hindcast and to spin-up model simulations from. Such calibration can increase confidence in glacier evolution models over a timescale that is representative to that of their projections.

The aims of this study are therefore to (a) quantify the past glacier extent and past ice surface of Greenland GICs during the Little Ice Age (LIA; 1900: Kjær et al., 2022) termination, and (b) make a critical analysis of the variability in mass loss rates by region and by glacier type. We achieve these aims by employing a novel large-scale methodology to calculate glacier mass changes over a longer-term (centennial) timescale since the LIA.

2. Data Sets and Methods

2.1. Little Ice Age Moraine Mapping

We mapped LIA extents for 5467 GICs larger than 1 km² in the RGI v6.0 inventory (mean year 2001; GLIMS, 2021) primarily using a hillshaded image of the ArcticDEM (Porter et al., 2018) with 2 m horizontal resolution. Reference was also made to color optical-wavelength satellite images from PlanetScope (Planet, 2021) and, where that was not available, from 15 m resolution ESRI ArcGIS World Imagery. Moreover, we aided our mapping using visual interpretations of landforms in the 2 m AeroDEM orthomosaic (Korsgaard et al., 2016), and using land-cover classifications of recently disturbed sediment made from 10 m resolution Sentinel 2 images from July and August 2021. Whilst the number of GICs larger than 1 km² is roughly one third of the ~17,000 GICs within the RGI v6.0 inventory (GLIMS, 2021) with no connection to the GrIS (Rastner et al., 2012), they account for 99% of total GIC area.

We mapped LIA glacier extents by extending the RGI v6.0 outlines (GLIMS, 2021) down-valley to the crests of prominent moraine ridges interpreted to represent the LIA termination. That interpretation was based on position, typically several kilometers down-valley from contemporary GICs, and morphology and surface character. Moraine ridges were often sharp-crested, steep sided, and typically devoid of vegetation (Figure 1a, Figures S2 and S3 in Supporting Information S1). If the LIA moraine comprised a set of multiple ridges, we deliberately mapped the innermost crest, which typically appeared more subdued in relief, more vegetated, and likely formed during older late-Holocene advances (Kjær et al., 2022). Outlines from LIA lateral moraine ridges maintains consistency, permitting replicability, and provides a conservative estimate of glacier extent during the LIA termination. We were unable to map LIA extents where geomorphological evidence was absent such as for some water-terminating GICs and for some outlet glaciers in the extreme north. Overall, this mapping protocol together with our observations of some empty cirques above the regional LIA ELA (cf. Carrivick et al., 2019) means that our analysis provides a minimum estimate of LIA area, and of volume loss and mass loss since the LIA. Our calculations of glacier areal extent changes are to 2001 as that is the mean date of images used to construct the RGI v6.0 glacier outlines.

2.2. Ice Surface Reconstruction

To mitigate artifacts and high uncertainty in parts of the DEMs, due to snow cover and cloud cover in the images used to make them, we limited our analysis of surface elevation changes to glacier ablation areas only. Glacier ablation areas (for LIA and RGI v6.0 glacier outlines) were delineated using the Area-Altitude Balance Ratio (AABR) method, which was automated using code developed by Pellitero et al. (2015) and as adapted and used across several other world regions (Carrivick et al., 2019; Carrivick, Andreassen, et al., 2022; Carrivick, Tweed, et al., 2020; Lee et al., 2021). We assumed an AABR of 2.24 ± 0.85 as suggested by Rea (2009) for arctic glaciers but the suitability of this value varies on a glacier-by-glacier basis. We therefore aggregate our ELAs spatially, and we do not infer climate change from our ELAs. A LIA glacier surface was interpolated between point elevations extracted from the ablation area outline (Figure S2 in Supporting Information S1).



Figure 1. (a) Example from Tasiilaq Fjord of prominent moraine ridges (dotted line) that mark the extent of a small glacier during the Little Ice Age (LIA) termination. (b) Example of LIA glacier extent mapping and semi-automated derivation of elevation changes in LIA ablation areas for the Uummannaq region of west Greenland.

Calculating the difference between our LIA surface and the ArcticDEM indicated the surface lowering that has occurred (Figure 1b). Some positive elevation changes occurred, either due to glacier surges, convex glacier surfaces, or artifacts in the DEM. These causes could not be objectively discriminated and so we replaced all these positive values using a void filling spatial interpolation technique to negate them. Surface lowering maps were converted to a volume change by summing the elevation changes per area of interest (glacier, region) and multiplying by grid cell size. The difference in volume change computed with/without positives amounted to <-10% by region (Table S1 in Supporting Information S1). Importantly, our glacier mass balance estimates derive from these calculations of volume change across ablation areas, not from the ELA estimates. The parts of the ArcticDEM mosaic covering Greenland GICs were constructed from images dating between 2007 and 2017 and with a mean date of 2015, which we use for our volume change rate calculations. We take the LIA termination as year 1900 following Kjeldsen et al. (2015), and whilst acknowledging regional differences in the timing of the LIA termination (1850–1965: see Section 5.3 in Kjær et al., 2022). Our rates of mass loss are most sensitive to the year chosen for the LIA termination.

More details of our data sets and mapping are given in Supporting Information S1 where we also show: (a) that the effect on our calculations of excluding RGI v6 glaciers $<1 \text{ km}^2$ is near-negligible, typically 2% for areal extent and <5% for volume; (b) our LIA surface interpolation routine, (c) the sensitivity of our results to the choice of an AABR value for ablation area estimation, and to a date for the LIA termination, (d) our conversions of volume to mass and mass balance, and (e) our categorization of glacier types.

3. Results

There has been considerable fragmentation of GICs since the LIA termination in the form of tributaries separating from trunk glaciers. The 5327 GICs that we mapped LIA outlines for now number 5467; a 2.6% increase in number. Regionally, the total glacier extent during the LIA termination was between 76,342 and 84,378 km², and





Figure 2. Equilibrium Line Altitude (ELA) and terminus elevation (Min. elevation) across Greenland for Glaciers and Ice Caps (GICs) during the Little Ice Age (LIA) termination (red bars) and 2015 (blue outline bars) by region north (NO), north west (NW), north east (NE), central west (CW), central east (CE), south west (SW), and south east (SE). The mapped ELA difference depicts the mean ELA difference for all glaciers per colored hexagon.

so there has been at least 4,120 km² (5.1%) to 4,554 km² (5.7%) areal extent loss between 1900 and 2001. Areal extent losses between the LIA termination and 2001 were at least 17%–19% in the central west (CW) region, and at least 11%, 10%, and 9% in the south east (SE), south west (SW) and north west (NW) regions, respectively, but only ~1% in the north (NO) region (Table S1 in Supporting Information S1).

GICs across Greenland have lost at least 528–646 km³ volume since the LIA termination to year 2015. The greatest proportion of volume loss (25%–32%) has come from the north east (NE) region, but that region has 36%–40% of the total LIA glacier area. In contrast the SE, SW, and NW regions have undergone a disproportionate volume loss since the LIA with 8%, 11%, and 11%, respectively, comprising just 5%, 7%, and 8% of the total glacier area, respectively.

The median elevation rise of glacier termini per region has decreased with higher latitude on the west coast from 228 m a.s.l. (± 12.5 m a.s.l.) in the SW to 169 m a.s.l. (± 19.4 m a.s.l.) in the NW (minimum elevation; Figure 2). On the east coast the SE region is an anomaly to this latitudinal pattern, having the lowest terminus elevation



Geophysical Research Letters



Figure 3. Comparison of long-term rates of glaciers and ice cap mass balance in glacier ablation areas since the Little Ice Age termination 1900 to 2015 (red) with short-term rates between 2000 and 2019 (blue), the latter using the data set of Hugonnet et al. (2021). Boxplots illustrate the control of terminus environment on longer-and shorter-term mass balance in glacier ablation areas.

rises of any region, aside from the NO. Nonetheless, the SE region has ELA changes that fit the latitudinal pattern (Figure 2). ELAs have risen by 99 m (\pm 11.8 m) in the CW region, 82 m (\pm 12.5 m) in SW, 72 m (\pm 8.6 m) in SE, 52 m (\pm 19.4) in NW, 51 m (\pm 10.0) in NE, 26 m (\pm 17.3) in central east and just 5 m (\pm 7.4 m) in the NO region (Figure 2). These regional medians reveal a higher rise in ELAs on the west of Greenland compared with those on the east coast and a N-S trend of decreasing ELA change with increasing latitude. However, there is considerable intra-region variability in ELA changes, particularly on the east coast (Figure 2).

Using an ice density of 850 kg m⁻³ (Huss, 2013), the mass balance of all Greenland GICs in ablation areas since the LIA termination to 2015 has been at least -0.18 to -0.22 m w.e. yr⁻¹. That rate is at least -4.0 m w.e. yr⁻¹ less than the -0.60 m w.e. yr⁻¹ for 2000 to 2019 that we derived for Greenland GIC ablation areas only from the data set of Hugonnet et al. (2021). Thus, we suggest a conservative estimate of a 3 times faster rate of mass loss between 2000 and 2019 than the mean rate since the LIA termination. If we assume no mass loss from LIA glacier accumulation areas and apply our mass changes glacier-wide, then we estimate that the mass balance rate since the LIA for all GICs was -0.054 to 0.066 m w.e. yr⁻¹, which shows that the rate of -0.37 to 0.45 m w.e. yr⁻¹ for 2000 to 2019 is an order of magnitude faster than the mean since LIA (Table S1 in Supporting Information S1).

Similarly, by examining regional glacier changes, we find that the Greenland-wide mean GIC mass balance rate over the past 20 years is at least twice as more negative as the LIA to 2015 mean. In the SW and NW regions the rate has become approximately three times more negative, from -0.3 to -0.76 and from -0.26 to -0.79 m w.e. yr⁻¹, respectively, and in the NO region the rate has become five times more negative from -0.11 to -0.57 m w.e. yr⁻¹ (Figure 3). We note that the NO region was highlighted by Khan et al. (2022) for its rapid acceleration of glacier mass loss from 2003 to 2009. These inter-regional patterns are the same for absolute volume (Figure S6 in Supporting Information S1), that is, without consideration of the contributing glacier area and time duration. The GIC mass loss equates to 1.24-1.52 mm SLE since the LIA termination, that is, between 0.011 and

 0.013 mm yr^{-1} . We compute that the SLE of GIC mass loss between 2000 and 2019 has been 0.08 mm yr}^{-1} using the data set of Hugonnet et al. (2021). For comparison, the mean annual mass loss of the GrIS over the last 20 years has been an order of magnitude more at 2.7 mm yr^{-1} (Muntjewerf et al., 2020).

Overall, the spatial pattern of GIC mass balance has transitioned to become more spatially heterogeneous between 2000 and 2019 compared with between the LIA termination and 2015 (Figure 3). Specifically, the range of mean mass balance in the three west regions has gone from -0.11 m w.e. yr^{-1} since the LIA termination to -0.28 m w.e. yr^{-1} during 2000–2019, and the range in the three east regions has gone from -0.18 to -0.52 m w.e. yr^{-1} (Figure 3). The NO region has experienced a -0.46 m w.e. yr^{-1} change in mass balance rate (Figure 3).

Marine-terminating GICs had the most negative mass balances since the LIA termination compared with lake- or land-terminating (Figure 3). However, during the period 2000 to 2019 lake-terminating GICs had the most negative mass balances (Figure 3). We find 24 of these lakes formed between the LIA moraine and the contemporary ice front, thus changing the terminus environment from land-to lake terminating. As a result, lake-terminating GICs have experienced the greatest change in mass balance regime since the LIA termination (Figure 3).

4. Discussion

4.1. Mass Balance

Studies concerned with Greenland GIC glacier evolution, or of isostatic effects of glacier mass loss for example, will find it useful to consider that a total volume of between 528 and 646 km³ has been lost since the LIA termination, equating to between 449 Gt and 549 Gt at a mean rate of 4.34 Gt yr⁻¹. That rate is one order of magnitude less than the 27 to 42 Gt yr⁻¹ reported by Khan et al. (2022) for 2003 to 2021 for GICs in north Greenland. It is also an order of magnitude less than the 75 to 100 Gt yr⁻¹ reported by Kjeldsen et al. (2015) for the entirety of the GrIS since year 1900. However, the areal extent and volume of the GrIS is 1.72×10^6 km² and 2.9×10^6 km³ the GrIS (Box et al., 2022; their Table 1), which are two and four orders of magnitude greater than 90 × 10³ km² and $11.8 \pm 3.7 \times 10^3$ km³, respectively, for all of the GICs combined (Millan et al., 2022). Therefore, although the absolute magnitude of mass loss (and hence global mean SLE contributions) from GICs has been approximately one twentieth of that of the GrIS since the year 1900, the mass loss from GICs is disproportionately large, given their combined size, when compared with mass loss from the GrIS. The greatest uncertainty in our rate(s) of change is the date used for the LIA termination. An earlier date (e.g., 1880) would make a slower rate of change (i.e., 3.7 Gt yr⁻¹; Table S3 in Supporting Information S1) and that would suggest a greater change in the rate of mass loss when compared with 2000–2019.

Numerical models of glacier evolution should become mindful of changing terminus environments. Given that we find 242 fewer marine terminating GICs in 2001 than during the LIA termination and that the number of lake-terminating GICs have increased slightly by 24 (Figure 3), we interpret that (a) many GICs that were marine-based during the LIA are now terminating on land, and (b) some GICs have developed ice-marginal lakes at their termini as those lakes have formed in the last few decades and inside LIA moraine ridges, for example, as shown across Greenland by How et al. (2021), Carrivick, How, et al. (2022). The influence of lake effects on glacier mass balance (e.g., Carrivick, Tweed, et al., 2020), in addition to climate forcing, could explain why the greatest changes of mass loss have occurred for lake-terminating glaciers, which is an association that we find exists to a greater or lesser extent in all Greenland regions (Figure 3).

4.2. ELAs

Absolute LIA ELAs tend to gain elevation with increasing latitude on the east coast but have quite similar median elevations per region on the west coast (Figure 2). This latitudinal pattern of LIA ELAs is counter to what would be expected with a sole control of colder air temperatures northwards and demonstrates that effective (solid) precipitation is very important (cf. McGrath et al., 2017 for Alaska), perhaps as important as air temperature, for controlling mass balance sensitivity to climate change across Greenland. Specifically, precipitation across Greenland is higher in the south due to warmer ocean waters, atmospheric temperatures and more evaporation (Box, 2002), thereby sustaining GIC accumulation zones at lower elevations in southern Greenland than in the north. There is also a continentality effect whereby LIA ELAs were higher inland than on the coast (Figure 2 inset), presumably reflecting similar precipitation gradients to those today (e.g., Taurisano et al., 2004).

Furthermore, topography and hypsometry (rather than solely glacier area) affects glacier response times (cf. Raper & Braithwaite, 2009; McGrath et al., 2017), especially considering that ice caps, which become more prevalent northwards (regions NW, NO, NE), have a top-heavy (skewed to higher normalized elevations) hypsometry that contrasts with that of valley glaciers (cf. Pfeffer et al., 2014).

Spatial heterogeneity in GIC changes (Figure 2) has recently been noted by Khan et al. (2022) in north and north-east Greenland, by Cooper et al. (2022) in central-east Greenland, and also by Brooks et al. (2022) for their sample of 42 glaciers in south Greenland, even though the latter excluded glaciers with confounding properties such as debris cover, and marine- or lake-termini. Brooks et al. (2022) attributed this spatial variability in glacier responses to local topographically-modified microclimates. Some regions have experienced increased precipitation that can have a stabilizing effect on glacier mass loss (Bjørk et al., 2018). Such spatial diversity in glacier mass balances and ELA shifts must cast considerable doubt on using a small sub-sample of glaciers to reconstruct the spatial variability of climate changes (cf. Carrivick & Brewer, 2004).

Some glacier-specific rises in ELA since the LIA termination to 2015 have been >150 m but region medians are quite modest; most are just a few tens of meters, suggesting GICs are not changing geometry in pace with climatic shifts; that is, climate change has progressed faster than glacier response time. Therefore, we suggest that GICs have committed ice losses, just as the GrIS has (Box et al., 2022). In the central and northern regions this could be due to glacier thermal regime; polythermal and fully-cold-based glaciers recede and thin much more slowly than temperate glaciers, becoming in disequilibrium with climate (see paragraph 11 of Irvine-Fynn et al., 2011). The NO region has the least dispersed histogram of ELAs and therefore has greatest proportion of glacier areal extent distributed at a similar elevation meaning that it is the most sensitive region to a rise in ELA.

4.3. Global Comparison

Centennial-scale rates of GIC volume and mass losses since the LIA have been determined for other World regions, for example, a doubling in rate of mass loss for Vatnajökull, Iceland (Hannesdóttir et al., 2015), Patagonia (Glasser et al., 2011) and the Southern Alps, New Zealand (Carrivick, James, et al., 2020), and a ten-fold increase across the Himalaya (Lee et al., 2021). These studies have conducted their analysis via an inventory-style mapping approach, as herein, which mitigates problems of sampling bias and glacier-specific conditions that weaken the link between glacier geometry changes and climate fluctuations, including; (a) glacier size; larger glaciers lose the most absolute volume (Figure S3 in Supporting Information S1) but typically lose the least volume proportional to their area, (b) terminus environment; marine- and lake-terminating lakes typically have higher mass loss rates than land-terminating glaciers (e.g., Carrivick, Sutherland, et al., 2022; Lee et al., 2021; Mallalieu et al., 2021), (c) glacier surface characteristics; predominantly debris-cover effects, and (d) glacier surges, which are a rapid translation of ice mass usually into an ablation area often in the form of a bulge or kinematic wave. The Greenland-wide trebling (at least) in rate of mass loss identified in this study since the LIA is therefore in accordance with these other studies, but the NO region of Greenland stands out for a five times increase in rate, which comparing our results with the results of Khan et al. (2022) has apparently occurred entirely in the last two decades.

5. Conclusions

Greenland GICs are under-studied in comparison to the GrIS, yet are very sensitive to climate change and are important contributors to global mean sea level. In this study we have determined glacier areal extent changes, reconstructed a LIA ice surface, and calculated ice volume changes since then within the ablation areas of Greenland GICs. We compute volume loss rates that are proportionally higher than areal extent loss rates, which demonstrates the importance of glacier thinning to overall mass loss, especially in northern regions where ice velocities are especially low and glacier termini retreat slowly. Comparison of mass loss rates between the LIA to 2015 and 2000–2019 reveals that glacier mass loss has accelerated over the past 20 years by at least two to three times for most regions and by five times for the NO region. We find that marine influences on GIC mass loss have diminished since the LIA and that proglacial ice-marginal lake effects on glacier mass balances have apparently increased. There has been considerable spatial variability in glacier-specific mass changes and ELA changes, which leads us to caution against inferring regional trends from analyses on only a small number of glaciers. Given the sparse and disparate nature of palaeo-climate proxy data across Greenland, numerical modeling of

19448007, 2023,

10, Downl

from

nups://agupubs.on/inenorary

/10.1029/2023GL103950 by Leed:

glacier evolution could make use of our data set to help unravel the complex interplay between regional climate change and local topographic controls on GIC mass loss since the LIA.

Data Availability Statement

GICs are available from RGI v6.0 obtained from GLIMS (2021) http://glims.colorado.edu/glacierdata/. GrIS catchments are available from Mouginot and Rignot (2019) https://doi.org/10.7280/D1WT11 and we extended these to include GICs. A Greenland coastline is available from Gerrish (2020) https://data.bas.ac.uk/full-record. php?id=GB/NERC/BAS/PDC/01439. Our LIA glacier outlines (ArcGIS 10.6.2 shapefile polygons), LIA glacier ablation areas (polygons), and LIA ice surface (.tif grid) and change in elevation (.tif grid) are openly available from the University of Leeds Data Repository https://doi.org/10.5518/1285.

References

Bjørk, A. A., Aagaard, S., Lütt, A., Khan, S. A., Box, J. E., Kjeldsen, K. K., et al. (2018). Changes in Greenland's peripheral glaciers linked to the North Atlantic Oscillation. *Nature Climate Change*, 8(1), 48–52. https://doi.org/10.1038/s41558-017-0029-1

- Box, J. E. (2002). Survey of Greenland instrumental temperature records: 1873–2001. International Journal of Climatology: A Journal of the Royal Meteorological Society, 22(15), 1829–1847. https://doi.org/10.1002/joc.852
- Box, J. E., Hubbard, A., Bahr, D. B., Colgan, W. T., Fettweis, X., Mankoff, K. D., et al. (2022). Greenland ice sheet climate disequilibrium and committed sea-level rise. *Nature Climate Change*, 12(9), 808–813. https://doi.org/10.1038/s41558-022-01441-2
- Brooks, J. P., Larocca, L. J., & Axford, Y. L. (2022). Little Ice Age climate in southernmost Greenland inferred from quantitative geospatial analyses of alpine glacier reconstructions. *Quaternary Science Reviews*, 293, 107701. https://doi.org/10.1016/j.quascirev.2022.107701

Carrivick, J. L., Andreassen, L. M., Nesje, A., & Yde, J. C. (2022). A reconstruction of Jostedalsbreen during the Little Ice Age and geometric changes to outlet glaciers since then. Quaternary Science Reviews, 284, 107501. https://doi.org/10.1016/j.quascirev.2022.107501

Carrivick, J. L., Boston, C. M., King, O., James, W. H., Quincey, D. J., Smith, M. W., et al. (2019). Accelerated volume loss in glacier ablation zones of NE Greenland, Little Ice Age to present. *Geophysical Research Letters*, 46(3), 1476–1484. https://doi.org/10.1029/2018gl081383

- Carrivick, J. L., & Brewer, T. R. (2004). Improving local estimations and regional trends of glacier equilibrium line altitudes. *Geografiska* Annaler: Series A Physical Geography, 86(1), 67–79. https://doi.org/10.1111/j.0435-3676.2004.00214.x
- Carrivick, J. L., How, P., Lea, J. M., Sutherland, J. L., Grimes, M., Tweed, F. S., et al. (2022). Ice-marginal proglacial lakes across Greenland: Present status and a possible future. *Geophysical Research Letters*, 49(12), e2022GL099276. https://doi.org/10.1029/2022gl099276
- Carrivick, J. L., James, W. H., Grimes, M., Sutherland, J. L., & Lorrey, A. M. (2020). Ice thickness and volume changes across the Southern Alps, New Zealand, from the little ice age to present. *Scientific Reports*, 10(1), 1–10. https://doi.org/10.1038/s41598-020-70276-8
- Carrivick, J. L., Sutherland, J. L., Huss, M., Purdie, H., Stringer, C. D., Grimes, M., et al. (2022). Coincident evolution of glaciers and ice-marginal proglacial lakes across the Southern Alps, New Zealand: Past, present and future. *Global and Planetary Change*, 211, 103792. https://doi. org/10.1016/j.gloplacha.2022.103792
- Carrivick, J. L., Tweed, F. S., Sutherland, J. L., & Mallalieu, J. (2020). Toward numerical modeling of interactions between ice-marginal proglacial lakes and glaciers. *Frontiers of Earth Science*, 8, 577068. https://doi.org/10.3389/feart.2020.577068
- Cooper, M. A., Lewińska, P., Smith, W. A., Hancock, E. R., Dowdeswell, J. A., & Rippin, D. M. (2022). Unravelling the long-term, locally heterogenous response of Greenland glaciers observed in archival photography. *The Cryosphere*, 16(6), 2449–2470. https://doi.org/10.5194/ tc-16-2449-2022
- Farinotti, D., Huss, M., Fürst, J. J., Landmann, J., Machguth, H., Maussion, F., & Pandit, A. (2019). A consensus estimate for the ice thickness distribution of all glaciers on Earth. *Nature Geoscience*, 12(3), 168–173. https://doi.org/10.1038/s41561-019-0300-3
- Gerrish, L. (2020). The coastline of Kalaallit Nunaat/Greenland available as a shapefile and geopackage, covering the main land and islands, with glacier fronts updated as of 2017. (Version 1.0) [Dataset]. UK Polar Data Centre, Natural Environment Research Council, UK Research & Innovation. Retrieved from https://data.bas.ac.uk/full-record.php?id=GB/NERC/BAS/PDC/01439
- Glasser, N. F., Harrison, S., Jansson, K. N., Anderson, K., & Cowley, A. (2011). Global sea-level contribution from the Patagonian Icefields since the Little Ice Age maximum. *Nature Geoscience*, 4(5), 303–307. https://doi.org/10.1038/ngco1122
- GLIMS. (2021). Global land ice measurements from space: Glacier database. Retrieved from http://glims.colorado.edu/glacierdata/
- Hannesdóttir, H., Björnsson, H., Pálsson, F., Aðalgeirsdóttir, G., & Guðmundsson, S. (2015). Variations of southeast Vatnajökull ice cap (Iceland) 1650–1900 and reconstruction of the glacier surface geometry at the Little Ice Age maximum. *Geografiska Annaler: Series A Physical Geog*raphy, 97(2), 237–264. https://doi.org/10.1111/geoa.12064

Hopwood, M. J., Carroll, D., Dunse, T., Hodson, A., Holding, J. M., Iriarte, J. L., et al. (2020). How does glacier discharge affect marine biogeochemistry and primary production in the Arctic? *The Cryosphere*, 14(4), 1347–1383. https://doi.org/10.5194/tc-14-1347-2020

- How, P., Messerli, A., Mätzler, E., Santoro, M., Wiesmann, A., Caduff, R., et al. (2021). Greenland-wide inventory of ice marginal lakes using a multi-method approach. Scientific Reports, 11(1), 1–13. https://doi.org/10.1038/s41598-021-83509-1
- Hugonnet, R., McNabb, R., Berthier, E., Menounos, B., Nuth, C., Girod, L., et al. (2021). Accelerated global glacier mass loss in the early twenty-first century. *Nature*, 592(7856), 726–731. https://doi.org/10.1038/s41586-021-03436-z
- Huss, M. (2013). Density assumptions for converting geodetic glacier volume change to mass change. *The Cryosphere*, 7(3), 877–887. https://doi.org/10.5194/tc-7-877-2013
- Irvine-Fynn, T. D., Hodson, A. J., Moorman, B. J., Vatne, G., & Hubbard, A. L. (2011). Polythermal glacier hydrology: A review. Reviews of Geophysics, 49(4), 38. https://doi.org/10.1029/2010rg000350
- Khan, S. A., Colgan, W., Neumann, T. A., Van Den Broeke, M. R., Brunt, K. M., Noël, B., et al. (2022). Accelerating ice loss from peripheral glaciers in North Greenland. *Geophysical Research Letters*, 49(12), e2022GL098915. https://doi.org/10.1029/2022gl098915
- Kjær, K. H., Bjørk, A. A., Kjeldsen, K. K., Hansen, E. S., Andresen, C. S., Siggaard-Andersen, M. L., et al. (2022). Glacier response to the Little Ice Age during the Neoglacial cooling in Greenland. *Earth-Science Reviews*, 227, 103984. https://doi.org/10.1016/j.earscirev.2022.103984

Kjeldsen, K. K., Korsgaard, N. J., Bjørk, A. A., Khan, S. A., Box, J. E., Funder, S., et al. (2015). Spatial and temporal distribution of mass loss from the Greenland Ice Sheet since AD 1900. *Nature*, 528(7582), 396–400. https://doi.org/10.1038/nature16183

Acknowledgments

We gratefully acknowledge receipt of funding from INTERACT for access to DMI Qaanaaq. MG was in receipt of a NERC PANORAMA DTP NE/L002574/1 PhD studentship whilst completing work for this study. JA was supported by FWF project P 35388. KKK acknowledges support from the Programme for monitoring of the Greenland ice sheet (PROMICE) at the Geological survey of Denmark and Greenland (GEUS). AAB acknowledges support from the Villum Foundation, Villum Young Investigator Grant 29456. We thank three reviewers for their comments that helped to improve the text.

- Korsgaard, N. J., Nuth, C., Khan, S. A., Kjeldsen, K. K., Bjørk, A. A., Schomacker, A., & Kjær, K. H. (2016). Digital elevation model and orthophotographs of Greenland based on aerial photographs from 1978–1987. *Scientific Data*, 3(1), 1–15. https://doi.org/10.1038/sdata.2016.32
- Larsen, N. K., Siggaard-Andersen, M. L., Bjørk, A. A., Kjeldsen, K. K., Ruter, A., Korsgaard, N. J., & Kjær, K. H. (2022). Holocene ice margin variations of the Greenland Ice Sheet and local glaciers around Sermilik Fjord, southeast Greenland. *Quaternary International*, 607, 10–21. https://doi.org/10.1016/j.quaint.2021.06.001
- Lee, E., Carrivick, J. L., Quincey, D. J., Cook, S. J., James, W. H., & Brown, L. E. (2021). Accelerated mass loss of Himalayan glaciers since the Little Ice Age. Scientific Reports, 11(1), 1–8. https://doi.org/10.1038/s41598-021-03805-8
- Machguth, H., Rastner, P., Bolch, T., Mölg, N., Sørensen, L. S., Aðalgeirsdottir, G., et al. (2013). The future sea-level rise contribution of Greenland's glaciers and ice caps. *Environmental Research Letters*, 8(2), 025005. https://doi.org/10.1088/1748-9326/8/2/025005
- Mallalieu, J., Carrivick, J. L., Quincey, D. J., & Raby, C. L. (2021). Ice-marginal lakes associated with enhanced recession of the Greenland Ice Sheet. Global and Planetary Change, 202, 103503. https://doi.org/10.1016/j.gloplacha.2021.103503
- McGrath, D., Sass, L., O'Neel, S., Arendt, A., & Kienholz, C. (2017). Hypsometric control on glacier mass balance sensitivity in Alaska and northwest Canada. *Earth's Future*, 5(3), 324–336. https://doi.org/10.1002/2016ef000479
- Millan, R., Mouginot, J., Rabatel, A., & Morlighem, M. (2022). Ice velocity and thickness of the world's glaciers. *Nature Geoscience*, 15(2), 124–129. https://doi.org/10.1038/s41561-021-00885-z

Mouginot, J., & Rignot, E. (2019). Glacier catchments/basins for the Greenland Ice Sheet [Dataset]. Dryad. https://doi.org/10.7280/D1WT11

- Muntjewerf, L., Petrini, M., Vizcaino, M., Ernani da Silva, C., Sellevold, R., Scherrenberg, M. D., et al. (2020). Greenland Ice Sheet contribution to 21st century sea level rise as simulated by the coupled CESM2. 1-CISM2. 1. Geophysical Research Letters, 47(9), e2019GL086836. https:// doi.org/10.1029/2019gl086836
- Noël, B., van de Berg, W. J., Lhermitte, S., Wouters, B., Machguth, H., Howat, I., et al. (2017). A tipping point in refreezing accelerates mass loss of Greenland's glaciers and ice caps. *Nature Communications*, 8(1), 1–8.
- Pellitero, R., Rea, B. R., Spagnolo, M., Bakke, J., Hughes, P., Ivy-Ochs, S., et al. (2015). A GIS tool for automatic calculation of glacier equilibrium-line altitudes. *Computers & Geosciences*, 82, 55–62. https://doi.org/10.1016/j.cageo.2015.05.005
- Pfeffer, W. T., Arendt, A. A., Bliss, A., Bolch, T., Cogley, J. G., Gardner, A. S., et al. (2014). The Randolph glacier inventory: A globally complete inventory of glaciers. *Journal of Glaciology*, 60(221), 537–552. https://doi.org/10.3189/2014jog13j176
- Planet. (2021). PlantScope. Retrieved from https://www.planet.com/
- Porter, C., Morin, P., Howat, I., Noh, M.-J., Bates, B., Peterman, K., et al. (2018). ArcticDEM, version 3, Harvard Dataverse, V1. https://doi. org/10.7910/DVN/OHHUKH
- Rantanen, M., Karpechko, A. Y., Lipponen, A., Nordling, K., Hyvärinen, O., Ruosteenoja, K., et al. (2022). The Arctic has warmed nearly four times faster than the globe since 1979. Communications Earth & Environment, 3(1), 1–10. https://doi.org/10.1038/s43247-022-00498-3
- Raper, S. C., & Braithwaite, R. J. (2009). Glacier volume response time and its links to climate and topography based on a conceptual model of glacier hypsometry. *The Cryosphere*, 3(2), 183–194. https://doi.org/10.5194/tc-3-183-2009
- Rastner, P., Bolch, T., Molg, N., Machguth, H., & Paul, F. (2012). The first complete glacier inventory for the whole of Greenland. *The Cryosphere*, 6(4), 1483–1495. https://doi.org/10.5194/tc-6-1483-2012
- Rea, B. R. (2009). Defining modern day Area-Altitude Balance Ratios (AABRs) and their use in glacier-climate reconstructions. *Quaternary Science Reviews*, 28(3–4), 237–248. https://doi.org/10.1016/j.quascirev.2008.10.011
- Siegert, M., Bacon, S., Barnes, D., Brooks, I., Burgess, H., Cottier, F., et al. (2020). The Arctic and the UK: Climate, research and engagement.
- Sugiyama, S., Kanna, N., Sakakibara, D., Ando, T., Asaji, I., Kondo, K., et al. (2021). Rapidly changing glaciers, ocean and coastal environments, and their impact on human society in the Qaanaaq region, Northwestern Greenland. *Polar Science*, 27, 100632. https://doi.org/10.1016/j. polar.2020.100632
- Taurisano, A., Billionøggild, C. E., & Karlsen, H. G. (2004). A century of climate variability and climate gradients from coast to ice sheet in West Greenland. Geografiska Annaler: Series A, Physical Geography, 86(2), 217–224. https://doi.org/10.1111/j.0435-3676.2004.00226.x

References From the Supporting Information

- Hock, R., de Woul, M., Radić, V., & Dyurgerov, M. (2009). Mountain glaciers and ice caps around Antarctica make a large sea-level rise contribution. *Geophysical Research Letters*, 36(7), L07501. https://doi.org/10.1029/2008gl037020
- Japan Aerospace Exploration Agency. (2021). ALOS world 3D 30 meter DEM. V3.2, Jan 2021. Distributed by OpenTopography. https://doi. org/10.5069/G94M92HB
- Oien, R. P., Rea, B. R., Spagnolo, M., Barr, I. D., & Bingham, R. G. (2022). Testing the area-altitude balance ratio (AABR) and accumulationarea ratio (AAR) methods of calculating glacier equilibrium-line altitudes. *Journal of Glaciology*, 68(268), 357–368. https://doi.org/10.1017/ jog.2021.100
- Rootes, C. M., & Clark, C. D. (2020). Glacial trimlines to identify former ice margins and subglacial thermal boundaries: A review and classification scheme for trimline expression. *Earth-Science Reviews*, 210, 103355. https://doi.org/10.1016/j.earscirev.2020.103355

19448007, 2023,